

Received 15 June 2023, accepted 1 July 2023, date of publication 1 August 2023, date of current version 31 August 2023. *Digital Object Identifier* 10.1109/ACCESS.2023.3300418

# **RESEARCH ARTICLE**

# **UXBIV: An Evaluation Framework for Business** Intelligence Visualization

# SHUQING LIU<sup>®1</sup>, (Member, IEEE), HANKUAN ZHANG<sup>®2,3</sup>, ZHENHUA YANG<sup>®2,4</sup>, JUN KONG<sup>2</sup>, LIMIN ZHANG<sup>5</sup>, AND CHUNHAI GAO<sup>6</sup>

<sup>1</sup>College of Intelligence, Tianjin University, Tianjin 300072, China

<sup>2</sup>Department of Computer Science, North Dakota State University, Fargo, ND 58102, USA

<sup>3</sup>JLL Technology Company Ltd., San Francisco, CA 94111, USA

<sup>4</sup>Department of Graphic Design, College of Design, Iowa State University, Ames, IA 50011, USA

<sup>5</sup>College of Business, North Dakota State University, Fargo, ND 58108, USA

<sup>6</sup>Faculty of Education, Shenzhen University, Shenzhen 518060, China

Corresponding author: Chunhai Gao (gaochunhai@szu.edu.cn)

This work was supported in part by the National Natural Science Foundation of China under Grant 61602339, in part by the China Scholarship Council Fund for the Postdoctoral Program under Grant 201506255098, and in part by the Stabilization Support Program for Shenzhen Higher Education Institutions under Grant 20220811005233001.

This work involved human subjects or animals in its research. Approval of all ethical and experimental procedures and protocols was granted by NDSU IRB under Protocol BA20113 ("An evaluation framework for business information visualization").

**ABSTRACT** Nowadays, visualization is becoming an essential part of data analysis. Business Intelligence Visualization (BIV) is a powerful tool that helps modern business flows faster and smoother than ever before. However, research on BIV evaluation is severely lacking; most evaluation studies for BIV focus on usability, which have limited aspects covered for customers' needs. The purpose of this research is to develop a UX framework that evaluates BIV (UXBIV), including decision-making experience and interactivity. First, we did a literature survey for good understanding of research progress in related fields, and established a conceptual framework. Second, we performed a case study that implemented this framework to demonstrate how our framework can be used in real business. Our analysis shows that this UXBIV framework is quite reasonable, and it can capture the differences among various BIV designs from the users' standpoints. This UXBIV framework can help design BIV and promote better decision-making on business affairs.

**INDEX TERMS** Business intelligence, visualization, evaluation, user experience, framework.

## **I. INTRODUCTION**

Business processes are one of the most important assets of organizations, because they determine the success or failure of the organization in global markets [76]. Business Intelligence (BI) system is becoming more needed by organizations to analyze data and make strategic decisions [36]. As a vital part of the modern BI systems emphasizing selfservice, visualization has been rising rapidly in the BI and analytics industry for the past few years. Business intelligence visualization (BIV) can assist business decision makers to extract information from data, facilitate business judgement and decision making, thereby improving organizational

The associate editor coordinating the review of this manuscript and approving it for publication was Claudio Loconsole.

performance and fostering modern business processes to be faster and smoother than ever before [89].

With the wide application of BIV in the business process, the demand for better visualization effects is stronger and stronger. A good UX can enhance the decision-making performance and strengthen the competitiveness of corporate. Thus, a human-centric evaluation of UX is obviously vital to BIV. However, according to our investigation, there exist some limitations in the research and application of BIV evaluation. First of all, systematic research on UX evaluation is still scarce in this field. A framework towards UX of BIV is much needed. Second, research on BIV evaluation mostly focuses on usability, lacking other valuable factors of user experience for the successful design of BIV. Third, the current research on evaluation of BIV rarely covers the decision-making experience; nevertheless, BIV values more on the role of visualization in promoting decisions. In a word, considering business needs, it is necessary to establish a new UX evaluation framework for BIV.

This research aims to develop a framework for assessing the user experience of BIV, named UXBIV [106]. First, we conduct an in-depth literature survey to investigate necessary aspects in BIV evaluations. Then, we propose an evaluation framework of BIV, which contains independent variables, dependent variables and corresponding measurements, and the user study design for UXBIV. Furthermore, we carry out an empirical case study to test this framework, and compare three visualization designs with the guidance of UXBIV. In the future, designers can systematically study user experience by referring to UXBIV and develop BIV design strategies.

The remainder of this paper is organized as follows. Section II presents the background and motivation for user experience of BIV. Section III describes in detail the evaluation framework of BIV. Section IV provides a case study using the proposed framework to evaluate three kinds of BIV designs. Section V discusses rationality of UXBIV, practical application of UXBIV and future directions of this research, whereas section VI concludes this paper.

#### **II. BACKGROUND AND MOTIVATION**

This section includes concepts related to BIV, user experience of BIV, and motivations.

#### A. CONCEPTS RELATED TO BIV

In the era of big data, visualization, a clear way of visual communication, has emerged rapidly. Visualization is an interdisciplinary field that summarizes and presents data with simple and easy-to-understand designs, so as to convey information clearly and effectively [24]. When discussing visualization, we have to address that data visualization and information visualization are two sides of this concept, and they are often used interchangeably. However, when investigating scientific grounds and concepts, the mind and behavior of users, such as interactivity and cognitive aspects, are more prominent in information visualization than in data visualization; developing and improving data processing techniques have been more common in data visualization than in information visualization" [52]. In this research, we mainly focus on user's perspective to assess the visualization of business intelligence, rather than techniques in computing or data-processing. Therefore, in the article, we mainly discuss information visualization.

Information Visualization is an important means to analyze and interpret a large amount of information. It uses computers to interactively display unstructured and non-geometric abstract data sets [74]. The function of information visualization is to provide people with a powerful analytic tool, so that people can make full use of their own visual and cognitive abilities to observe and analyze information, thus discover the relationship patterns of information [74]. Currently, information visualization technology has been widely applied in Internet, medicine, biology, industry, agriculture, military affairs, political relations, entertainment information and business information [10], [21].

Business intelligence (BI) is the process of collecting, managing and analyzing business information, and its purpose is to promote the decision-making of enterprises [20], [25]. Business intelligence has been widely used in banking, insurance, securities and retail industries. BI suppliers vigorously promote their visualization functions, which prove the importance of BIV to modern organizations, since BIV is considered to be the core component of business intelligence [9], [72]. BIV uses computer-supported interactive visual representations to shows the complex relationships, potential information and development trends among original multidimensional business data, which promote better understanding of data, business, and behavior and enhance the insight of decision making on business processes [9]. Besides usability, aesthetics, pleasure and interactivity, the most important thing for BIV design is to provide decision support. Whether BIV can better reflect this design philosophy should be examined by user experience.

#### **B. USER EXPERIENCE OF BIV**

User experience has a long history that can be tracked back to late 1800s or early 1900s. The term UX was brought to wider knowledge by Donald Norman in the mid-1990s [75]. UX involves all aspects of users' interaction with a product or service [4]. ISO 9241–210 defines UX as the users' perceptions and responses when interacting with a product, system or service [44]. Users' demands, subjective evaluation and emotional feelings during this process are considered as the core of positive experience [55]. Providing excellent user experience can at least prevent the loss of existing users [45]. As people's demands continue to escalate, UX has gradually become a crucial factor for product or service success, and has also become a hot issue in the field of HCI, design and business [77]. Existing research indicates that it is necessary to define the corresponding elements or dimensions of UX according to the objects to be investigated [45].

In visualization field, there are several research conducted about visualization UX. These studies covered usability, aesthetics, interactivity, user tasks, etc. Usability can be described as the capacity of a system to provide a condition for its users to perform the tasks safely, effectively, and efficiently while enjoying the experience [58]. Stefania Passera has concluded visualization helped improving usability on contracts. Helena Dudycz has validated the usability of visualization as it pertains to semantic searches in the analysis of economic and financial indicators [22]. Khawaja et al. have researched how to measure cognitive load in behave for usability evaluation [51]. In 2006, De Angeli et al. proposed that not only usability is important to UX, but also interaction and aesthetics [19]. Wright et al. proposed a guideline about the aesthetics and UX-centered design [100]. And in 2009, Filonik and Baur have discussed how to measure aesthetics in information visualization [29]. Some other research has been conducted in interaction and interactivity, such as the survey proposed in 2011 by Khan and Khan [49]. Sherry Koshman has conducted a study about user interaction in visualization system which helps to understand better the novice/expert paradigm when testing a visualized interface design for information retrieval [54]. Buja et al. have discussed about interactions on high-dimensional data visualization [13]. In order to perform a study, the user task is also an important component in experimental design. Amar and Stasko proposed a task-based framework for evaluation of information visualizations [6]. Lee et al have performed taxonomy about tasks in graph visualization which concluded from various research [58]. Yi et al. also performed research to analyze interactions in information visualization [101]. Even there are such fields being researched, while we focus on the UX, we find that most of the research only cover usability. Saket et al. also noticed this in 2016 and proposed several other important aspects for UX in visualization [83]. Thus, we believe it is necessary to expand UX evaluation in BIV to somewhere beyond usability.

Another emerging field related to BIV is dashboard evaluation. Dashboard is only a subset of BIV that includes the broader scope of displaying and analyzing various data visually to support decision-making. Some of the dashboard research focus on usability: Magdalena et al. have presented a strategy to increase the utilization of BI dashboards by conducting user testing and heuristic evaluation, their evaluation includes learnability, memorability, efficiency, error detections, and user's satisfaction [107]. Almasi et al. emphasized that a framework based on usability principles is necessary, proposes specific dashboard usability criteria (such as usefulness, operability, learnability, etc.) and identifies commonly used evaluation questionnaires (such as System Usability Scale (SUS), Technology Acceptance Model (TAM), Situation Awareness Rating Technique (SART), etc.) [109]. Orlando and Sunindyo proposed a flow to development and modification of a BI dashboard and evaluate the usability with SUS and Usability Metric for User Experience (UMUX) [110]. From a starting point of usability, some researchers have shifted their focus towards a broader concept of UX: Muppidi et al. stated that user experience is critical to the development of successful business intelligence solutions, proposes a new UX model for decision-making in the design and development of BI dashboards. They use 16 UX factors to evaluate dashboard design such as operability, memorability, satisfaction, etc. [108]. Burnay et al. proposed a review of existing BI modeling notations and a framework of dashboard, their UX evaluation focus on efficiency and expressiveness [112]. However, there are several limitations in existing dashboard evaluation studies. Firstly, dashboards are a subset or specific application of visualization, so dashboard evaluation cannot substitute for the evaluation of the entire visualization in BI. Secondly, few studies have examined decision-making experience, which is a crucial aspect of user experience in BIV. Thirdly, current research on evaluation has not summarized independent variables and decision-making tasks, which have overlooked the perspectives of designers and developers, and has not provided convenience for manipulating design features. Furthermore, previous research has not yet conducted empirical analysis to examine the validity and reliability of the evaluation framework of BIV.

In order to improve the productivity and efficiency of enterprises, user experience of BIV has become a new field of increasing interest to scholars. Although there are few existing studies on UX of BIV, this field is attempting to facilitate humans' interactions with visualization and to develop easyto-use visual intelligent systems for decision-making [8]. However, current UX studies about BIV mainly focus on usability evaluation. Specifically, Chung and Leung [16] compared a visualization prototype (SNV) with traditional method (Web browsing and searching) on the analysis of business stakeholder information. Results showed that the information presented on SNV was more useful for analyzing than on the Web site, and SNV was perceived to be more capable in helping effective analysis and decision-making. Yun et al. [102] developed a novel visual decision support system (DSS) based on different data-mining techniques. The indicator of evaluation is the number counts of user's positive or negative evaluation on this system. Researchers collected the users' brief comments on this system. The results indicated that Concept Lattice-based method achieves the best performance, since this method received the highest positive evaluation rate among all the methods. Ltifi et al. [66] tried to combine visualizations with data-mining techniques to promote decision making in a newly developed visual intelligent decision support system (VIDSS). The result of user study demonstrated that VIDSS have good rating scores on usability. Basole et al. [12] evaluated the usability and usefulness of three visualization methods (list, matrix, network) for ecosystem analysis. List was considered the easiest to learn but the least useful. Network was rated the highest in virtually all ratings. Matrix was rated the most difficult to learn, but relatively useful for ecosystem analysis. Bačić and Fadlalla [9] proposed some BIV elements suggested as independent variables for UX studies on BIV, which expanded research ideas in this area. Their study presented BIV elements according to visual mental abilities in Non-Verbal Intelligence Quotient (NVIQ): exploration, interaction, business acumen and relevant data, analytics and statistics, representation, perception, cognition, cognitive effort, memory and storytelling. All these BIV elements should be regarded as significant factors that affect decision-making performance. However, for designers and UX researchers, if these BIV elements are considered as independent variables, they are too abstract and difficult to manipulate, and the design problems cannot be identified directly and quickly.

# C. MOTIVATION

First of all, according to Maslow's hierarchy of needs, levels of needs are constantly improving, so users will not just stick to basic functions, but also pursue a satisfying experience [103]. The business success of products depends more and more on a pleasant user experience [55]. Most users of BIV do not have enough technical background and do not know the technical details of the system, so the quality of user experience about BIV system can have a critical impact on the work efficiency of decision makers. A satisfactory user experience provided by the BIV system can boost work performances of decision makers and bring obvious competitive advantages for development companies.

Second, research on BIV evaluation mostly focuses on usability. For example, Amyrotos noticed the importance of UX in BIV and proposed a user-centered framework, but his research focuses on usability such as effectiveness and accuracy [111]. However, good usability is not enough to create a good UX, because usability is only one part of the user experience. User experience includes all aspects of users' interaction with products or services. The dimensions of UX contain pragmatic aspects and hedonic aspects [38], that is, from traditional usability to aesthetics, appeal, and pleasure, etc. [1], [55]. Users' needs and their subjective and emotional evaluation of interaction are considered as the core of positive experience [55]. UX should also customize elements according to the characteristics of products or services [45]. UX of BIV lacks many valuable factors of user experience, such as trust, aesthetics, emotion, interactivity, loyalty etc., which are very important for the successful design of BIV.

Thirdly, the current research on BIV evaluation rarely covers decision performance. The existing BIV research indicators of decision-making performance are only the accuracy and speed of decision-making tasks, but not involve decision difficulty, time pressure, perceived information quality, decision-making quality, decision confidence, satisfaction and so on [42], [97]; In addition, decision-making style of users will also affect the decision-making performance in BIV [2]. All these factors can help BIV design is more dedicated. The main goal of BIV is to nudge business decisionmaking. Compared with visualization in other fields, BIV values more on the role of visualization in promoting decisions. Thus, it is essential to assess the decision-making experience in the fields of BIV evaluation.

According to the above analysis, a UX evaluation framework for BIV is urgently required. This structured framework should basically contain the important components of UX research and relationships among them. By using UXBIV, designers and developers of BIV can know how to study user experience of BIV. The independent variables include factors or elements that may affect the UX of BIV; the dependent variables involve dimensions of UX reflect the changes of independent variables; The framework also comprises user study design (selected tasks, methods and a paradigm) for use experience of BIV. Meanwhile, we should also provide a case study based on UXBIV to promote the understanding and application of this framework. We hope that UXBIV can meet the needs of intelligent analysis industry, promote decision performance and customer loyalty, and

#### TABLE 1. Articles reviewed based on interested areas (contains overlap).

Visualization Techniques	Business Intelligence	Visualization Design	UX I	Evaluatior	Framework	Survey
115	30	20	51	43	13	12

enhance the competitiveness and influence of BI development companies.

#### **III. FRAMEWORK OF UX EVALUATION FOR BIV**

This section includes the evaluation methodology for BIV, the independent and dependent variables of the evaluation for BIV, and the user study design of the evaluation for BIV.

# A. METHODOLOGY

The development of UXBIV framework came in three different steps through literature study:

- *Search phase:* we have obtained various of articles through Web of Science and Google Scholar, based on the scope of BIV, UX, decision-making, and visualization evaluation (e.g., visualization AND user experience, visualization AND evaluation, visualization OR evaluation AND framework, etc.). We have found 287 articles at the end of this phase, including 177 from Google Scholar, and 110 from Web of Science.
- *Screen phase:* when an article has been found, we conducted a screening on the article, and categorize it into several different interested areas based on their research goals. In this phase we have removed 146 articles that not very related to our topic, and pushed 141 articles into Identifying phase.
- *Identifying:* we further processed these articles, and label all other interested areas covered by the articles.

After these steps, we have gained a set of literatures (Table 1) that can be used to build our framework:

Findings from this literature study are used to create UXBIV evaluation framework. This framework compiled from all the interested areas based on existing research. The structure of UXBIV is shown in Figure.3, and we will introduce our findings in the following sections.

#### **B. INDEPENDENT VARIABLE OF BIV EVALUATION**

According to literature analysis, we have identified 4 main aspects that have influences to user's experience, and two smaller areas that have fewer impacts as side-aspects. Figure 1 shows the independent variables (IV) in UXBIV.

# 1) USER'S BACKGROUNDS

User's background is a widely used metric in surveys. In order to evaluate user experience, background information must be considered, and we can use related questionnaires to measure this information [51], [56]. Two kinds of backgrounds are being taken into consideration into our framework: Physical background and experience background. Physical background is referring to the demographic features of users,



FIGURE 1. Independent variables (IV) in UXBIV.

such as age, gender, race, etc. These features will provide us statistical evidence about whether a design related to user's biology traits [32], [61]. Experience background refers to user's experience and history, like professions, computing device usage, familiarity to data, etc. A well-trained person will have higher efficiency and accuracy, and may require a better design to meet their expectations [57], [68].

# 2) DATA CHARACTERISTICS

Accurate handling of data is essential to ensure the performance and reliability of the system, and any errors can have significant impacts [50]. Based on the survey we have conducted, each chart type has its own pros and cons. In addition, some design elements, such as legend and label, may help extend the usage of a chart type to cover different data types. Thus, we have to consider the mapping between data characters and design since this is directly connected with user experience. These are objective measures that can be computed from a dataset.

- *Time Dependence:* A data set can be either related to time or not, which can be easily categorized into time-dependent data or non-time-dependent data. It is important to choose the proper format for time-dependent series visualization. Time-dependent data usually use line chart or bar chart; in BI it is usually used in time-dependent reports such as sale report [39]. Non-time-dependent data such as warehouse coverage is the kind of data which are not linked to time, and usually we use map, heat-map, or tree chart to render such kind of data [73].
- *Dimensions:* Dimension refers to how many attributes a dataset has [30]. Commonly used dimensions are people, products, place and time. High Dimensional data are difficult to calculate and visualize. Specialized visualization techniques might be necessary to transfer high-dimensional data into visualizations [13], [87]. Thus, we believe these techniques needs to be evaluated because different visualization approach may affect UX.
- *Volume:* Volume refers to the data size or scale [98]. The data processing and visualization with large volume of data is also known as "big data" in modern research. In order to visualize large volume of data within a limited displaying scope, specialized techniques are necessary [35], [81], [98]. Considering different visualization techniques may result in different visualization elements displayed, the UX might be affected, we need to take the data volume and techniques into consideration of BIV evaluation.

### 3) INTERACTIONS

Interaction refers to all the tasks that the user can perform to interact with the visualization interface to retrieve information they want [101]. It includes but not limited to button, drag tool, zoom tool, color tool, etc. We can compare different interactions used in designs: the more interactions provided will grant user more degrees of freedom, but more interaction elements might reduce the usability of the visualization [28], [84]. Therefore, the relationship between interactions and UX should be explored in visualizations.

# 4) DESIGNS

The difference in design may lead to direct impact towards user's experience. Choosing the diverse type of visualization would result in difference in users' feedback [40]. Thus, we have to think about interface design. We can count different design types based on the following aspects:

• *Style*: Style is an abstract concept that relates to how an artefact, such as visualization, can be recognized and be potentially grouped in a specific category [71]. Some

empirical evidence exists that style plays an important role in the perception of users, as it is often the only 'way' to make a product stand out [93]. Moere et al. [71] have categorized style into 3: analytical style, magazine style, and artistic style. These styling have difference in many distinct dimensions, bringing diverse overall visual representatives and affecting UX in an upper level of design.

- *Organization*: Organization refers to the layout and arrangement of design elements in visualization [2]. Different approaches of information organization can affect users' cognitive load, interpretation, performance and user satisfaction [2]
- *Chart Type:* Chart Type refers to the visualization types such as line chart, bar chart, pie chart, etc. Various chart type leads to different user's experience [18]. UX of Chart Type is still a key component to be discussed in visualizations [31], [33], [70].
- *Aesthetic Element:* Aesthetic elements refer to visual elements such as color, color blending, font, text size, textures, layout, etc. Well-designed aesthetics will help BIV to improve user experience [11], [14], [39].

# 5) CHARACTERISTIC FACTORS AND ENVIRONMENTAL FACTORS

Characteristic factors and environmental factors may have direct impact on user experience, so we can consider these factors as independent variables in UXBIV. In some cases, although we do not take these factors as independent variables, it is necessary to set them as controlled variables, to ensure the reliability of the research.

In our framework, characteristic factors refer to distinguishing features of individuals' personality that may affect user experience; we can use related questionnaires to measure this information. Some dimensions which might fall into this category is: decision styles (directive, analytical, conceptual, and behavioral), time perception (monochronic/polychronic), high-context/low-context, long-term/short-term orientations, thinking styles (holistic/analytic thoughts) and uncertainty avoidance [2], [95].

Environmental factors are related to the surroundings of users, including two aspects: the social backgrounds (cultural, technological, economical, etc.) and the experiment contexts (lab, online, office, etc.). Social background, which brings different thinking path, affects user's response [43]. Besides, experiment context may influence users' performance: users felt more confident in testing in a professional lab than online, and the data collect in the first environment was more reliable [15], [41]. We can count different types of environment factors that user may involves.

#### C. DEPENDENT VARIABLES AND MEASUREMENT

Based on the literature analysis, we summarized the dependent variables (DV) in UXBIV. Dependent variables of UX can be based on tasks or overall impressions. The task-based UX can be rated by users at the end of each decision-making task. Meanwhile, the overall UX is usually evaluated immediately after participants complete all the interactions with BIV. Figure 2 shows the dependent variables (DV) in UXBIV.

# 1) TASK-BASED DEPENDENT VARIABLES

- a: OBJECTIVE MEASURES
  - *Reaction Time and Accuracy*: Reaction time and accuracy are two widely used research measures for humancentered research. Reaction time (RT) refers the time spent of a user to complete a specific task. RT be used to evaluate usability and cognitive loads in user study [82], [83]. In our research, the accuracy is the correctness of the user to complete specific tasks. This can be used to track accessibility and cognitive loads [34], [80].
  - *Electrophysiology*: Electrophysiology is another powerful method of collecting objective data. Some representative electrophysiology includes eye-tracking, electroencephalography (EEG), electrocardiography (ECG) and electrodermal activity (EDA), etc. Electrophysiological techniques can measure user experience in real time, such as attention, cognitive load and emotional state in the process of human-computer interaction, and facilitate identifying the influence of graph design towards UX [3], [34]. The combination of subjective and objective evaluation can make a more comprehensive and convincing evaluation of user experience. In our framework, this kind of measure is not mandatory, but it is a great addition to help us improving the evaluation.

# **b:** SUBJECTIVE MEASURES

UX is a term that related to user's subjective feedback, so we should also use subjective measures to analyze UX [78]. The task-based UX can be rated by users immediately after each decision-making task. This kind of measures is direct to user's workload and satisfaction of tasks [96].

One common measure is After-Scenario Questionnaire (ASQ) [60]. A total of 3 items are incorporated into 3 different dimensions: effectiveness, efficiency, and satisfaction. All items can be rated on a 7-point scale. Another popular measure is NASA Task Load Index (TLX) [37]. It allows users to perform subjective workload assessments on various human-machine interface systems. TLX measure on 6 different dimensions: mental demand, physical demand, temporal demand, performance, effort, and frustration. All items can be rated on a 7 -point scale.

There are also more measures available that we can use for subjective measures. Depends on different focus of the actual evaluation taken, we could have flexibility to choose different subjective measures.

# 2) OVERALL UX AS DEPENDENT VARIABLES

According to literature analysis, Overall UX of BIV should contain three major aspects: attractiveness, decision-making experience and interactivity. The overall UX is usually evaluated immediately after participants complete all the interactions with BIV.

- *Attractiveness*: Based on the concept of user experience, BIV attractiveness comprises a set of UX aspects which measure the traditional comprehensive impression of the evaluated BIV system. BIV attractiveness consist of usability, trust, appearances, emotional involvement and loyalty. These ingredients can be measured with Standardized User Experience Percentile Rank Questionnaire (SUPR-Q) [85] and Personal Involvement Inventory [104].
- *Usability*: Usability refers to the effectiveness, efficiency, safety, utility, and learnability of a BIV design [12]. A well-designed BIV should provide sufficient information and make it easy to use. Thus, usability test is about to measure how easy-to-use of a system from user's perspective. For example, if the user think one design is easier to use than another, the first one should have higher rating on this metric.
- *Trust*: In UXBIV, trust is defined as attitudes held by a user derived from his or her perceptions about certain the brand, products or services. Trust is not only linked to relationships within a business environment [23], but also plays a strong role in human-computer interaction [47]. Trust is a "mediating factor" that determines whether consumers accept and use an automated system for self-service [59].
- *Appearance*: Appearance is a metric that measures aesthetic appeal of BIV design. This metric evaluates user's feedback based on design appearance. Cawthon [14] thinks that visualization attractiveness is an important part of user experience. For example, a design with well-designed color palette should have better visual feedback than a design with only black and white, and the rating should be higher.
- *Loyalty*: Loyalty refers to users' strong commitments and attachments toward products or services [27]. Loyal users have more confidence about re-using BIV system, and/or the willingness to share with others. Furthermore, loyal users are not easily distracted to slightly more attractive alternatives. According to Torres-Moraga et al. [92], loyalty combines high attitude orientation and repeated purchase behavior. Ratings of loyalty directly demonstrates the user's satisfaction about the system.
- *Emotional Involvement*: During the interaction with the system, users also experience various feelings including positive and negative emotions [99]. Emotional involvement measures the emotions, moods and feelings evoked by interacting with systems [46]. Emotional Involvement Scale (EIS) is Seven-point Semantic Differential Scale with a set of adjectives to let the user rate the system (e.g., boring/ interesting).

# 3) DECISION-MAKING EXPERIENCE

Decision-making is a crucial component to be considered while designing BIV. Thus, in our framework, we will evaluate it from four aspects: decision complexity, auxiliary



FIGURE 2. Dependent variables (DV) in UXBIV.

importance, decision quality, and information quality. The measurements of these four aspects come from the Decision-making Quality Questionnaire (DQQ) published by Visinescu, Jones and Sidorova in 2016 [97].

• *Decision Complexity*: Decision complexity is defined according to the number and kinds of variables or elements involved in decision-making and their interactions [53], [97]. Decisions with more alternatives and

attributes were evaluated by decision makers as more difficult [91]. A decision that needs to consider more elements and their relationships will be rated higher on complexity [97].

- Auxiliary Importance: In UXBIV, Auxiliary Importance evaluates how important the visualization is during the decision-making process. This dimension depends on the usage level of BIV, which refers to the extent to which users use and rely on BIV to make decisions [17], [97]. If decision-making relies more on the BIV system, the rating of this system should be higher.
- *Information Quality*: Tractinsky and Meyer [94] have defined this as "relative efficacy to provide the relevant information for the viewer". In UXBIV, information quality is measured by two dimensions: representational (e.g., understandability) and accessibility (e.g., ease of use) [97]. For example, a design that gives users hundreds of data points may lead to difficult to understand and the user will suspect the information retrieved may be wrong. In this case, the Information Quality rating should be lower.
- *Decision Quality*: Decision quality is often reflected by decision performance, such as decision-making accuracy and speed. However, Subjective measurement is also indispensable and significant. The perceived decision quality can be considered as the perception of the decision results, which can be measured by perceived satisfaction, confidence and trust of the outcomes in decision-making process [97].

# 4) INTERACTIVITY

In UXBIV, we propose to use interactivity as an indicator of interaction. Interactivity refers to the degree to which two or more communication parties can act on each other on the interaction system, the information and synchronization [64]. Interactivity can be evaluated from three aspects: perceived control, perceived responsiveness, and perceived communication.

- *Perceived Control:* Perceived control refers to the ability to manipulate the information flow [7]. This metric evaluates the satisfaction of human operation. If the system provides enough control elements to let the user can choose the timing, content, and sequence of a communication, it should be rated higher on perceived control. This metric can be measured by Interactivity Perceptions Scale (IPS) [63], [69].
- *Perceived Responsiveness:* Perceived responsiveness refers to the degree to which the responses in a communication are perceived to be appropriate and relevant, and resolving the information need of the interaction episode or event [48]. The system response should follow user's expectation and provide desired information. If the user does not get what they want after performing an input to the system, the satisfaction about the system will be

decreased, and this rating will be lower. This metric can also be measured by IPS [63], [69].

• *Perceived Communication:* Perceived Communication is defined as the extent to which users believe that the system facilitates two-way communication [86]. This metric evaluates the satisfaction of communication reciprocity between human and systems. A good perceived communication offers individuals active control and allows them to communicate both reciprocally and synchronously. This metric can also be measured by IPS [63], [69].

# D. USER STUDY DESIGN

User study based on behavioral science is the chance to "gather information about the users' performance with the system, their comments as they operate it, their post-test reactions and the evaluator's observations" [67]. The section of user study design in UXBIV include decision-making tasks, methods and paradigm (Figure 3).

# 1) TASK

In this section we will introduce a guideline to design evaluation tasks for the user study. Based on literature analysis, with the consideration of daily usage in business, we come up with a task set that would fits business visualization better.

# a: MINIMAL TASK CATEGORY

In Table 2, we list the tasks of visual evaluation in the core literature. In Table 3 we summarize the minimal task unit commonly used in BIV evaluation and provide examples of each task.

Since we're focusing on BIV, with this novel businessrelated categorization of tasks, we can create propriate tasks for BIV design analysis. For example, with a given warehouse dataset, a normal use case is to retrieve sales data based on time or destination. These tasks can be easily built by combining Precise select/ Compare/ Rank/ Cluster/ etc. depends on the interaction/ visualization technique used.

# b: COMBINED TASKS

Lee et al. [58] noticed several tasks above low-level tasks. Similarly, we can also develop some combined tasks. However, as there are so many possibilities for the combinations have so many possibilities, here we will only provide some examples.

Connectivity: This is a combination with precise/ fuzzy select, compare, and correlate. For example, user finds ware-houses with the largest demands, and links them together for a best route to resupply.

Strategy making: This is a combination with fuzzy select, cluster, and estimate. For example, the user uses heat-map to read sales in different areas, and make estimation for next season.



#### 2) USER-BASED METHODS AND PARADIGM

As viewed from application, how to evaluate user experience is a very important issue. Similar to the traditional UX research, there are various user-based methods for UXBIV. Qualitative methods include interview, focus group, and thinking-aloud; quantitative methods include questionnaire, behavioral experiment and electrophysiology [55], [105]. These methods can be combined in an evaluation of user experience. The quantitative data collection can be obtained by automated tools; meanwhile, the combination of qualitative research methods such as interviews and focus groups is expected to enable evaluators to have a deeper understanding of user experience and optimize BIV design through user feedback [55], [79], [105].

In addition, enlightened by Thayer and Dugan [90], we also provide a paradigm of user test for BIV, which includes all important phases: (i) Participants enter the research site, and researchers introduce the research content; (ii) Participants read the informed consent form and sign it; (iii) Participants receive background survey; (iv) Participants complete the experiment training with the aid of researchers; (v) In the formal experiment, every time the subjects complete a task, they immediately evaluate this task; (vi) After all the in-task surveys on each design, participants complete the post-study survey on each design; (vii) Researcher conducts the interview study; (viii) Participants have a short break; (ix) The sequence of steps from (v) to (vii) can be repeated several times; (x) After interview, participants are acknowledged and given compensation. Figure 4 shows the stages of whole user test and corresponding methods.

# **IV. CASE STUDY**

The main purpose of the user study was to apply our evaluation framework to an actual evaluation case. This demonstrated how to use our framework in the BI visualizations that we developed. Besides, we also assess the diverse designs



**FIGURE 4.** Combination of the Paradigm and corresponding methods.

of the given BIV system by user-centralized evaluation and make design decisions.

# A. PARTICIPANTS

We used on-campus online advertisements to recruit users that are familiar with business visualization systems. 47 eligible participants who can speak fluent English were recruited from a university in the United States. All the participants have normal vision or corrected vision. 84% of all 47 participants are male, and 16% are female. The subjects are from different majors (62% of computer science, 17% of management information systems, 9% of mathematics major,

#### TABLE 2. Task categories in previous research.

Author	Year	Tasks
		Identify: looking for some data with
		characteristics.
		Locate: Find something and deliver its relatives.
~		Filter: Find directory containing files of
Stasko et al.	2000	particular type.
[88]		Compare: Compare size of two files and identify
		the larger.
		Find duplicate: Find identical elements
		Compare cluster: Compare two directories.
		Retrieve Value: Given a set of cases find
		attributes of those cases
		Filter: Given some conditions on attributes
		values find data cases satisfying those
		conditions
		Compute Derived Value: Given a set of data
		cases compute an aggregate numeric
		representation of those data cases
		Find Extremum: Find data cases possessing an
		extreme value of an attribute over its range
		within the data set
		Sort: Given a set of data cases rank them
		according to some ordinal metric
		Determine Range: Given a set of data cases and
Amar and	2005	an attribute of interest, find the span of values
Stasko [5]	2005	within the set
		Characterize Distribution: Given a set of data
		cases and a quantitative attribute of interest
		characterize the distribution of that attribute's
		values over the set
		Find Anomalies: Identify any anomalies within
		a given set of data cases with respect to a given
		relationship or expectation e.g. statistical
		outliers
		Cluster: Given a set of data cases find clusters
		of similar attribute values
		Correlate: Given a set of data cases and two
		attributes determine useful relationships
		between the values of those attributes
		Scan: Quickly review the list of items requires
		users to review many items at once but not
		necessarily to retrieve exact values
Lee et al. [58]	2006	Set Operation: Given multiple sets of nodes
		perform set operations on them. For example
		find the intersection of the set of nodes
		Select: Mark a data item(s) of interest to keen
		track of interest
		Explore: Examine a different subset of data
		case
		Reconfigure: Provide users with different
		perspectives onto the data set by changing the
		spatial arrangement of representations
		Encode: Alter the fundamental visual
		representation of the data including visual
Yi et al. [101]	2007	appearance (e.g., color, size, and shape) of each
		data element.
		Abstract/Elaborate: Adjust the level of
		abstraction of a data representation.
		Filter: Change the set of data items being
		presented based on some specific conditions.
		Connect: (1) highlight associations and
		relationships between data items that are already
		represented and (2) show hidden data items that
		are relevant to a specified item.
-		Find Extremum of change: Localizing the
		greatest increase or greatest decrease subjective
		evaluation of user experience in interactive 3D-
		visualization in a medical context.
Livingston		Counting: How many strings are in the scene
and Decker	2012	Find Extremum: Which string is the closest to
[65]		vou.
		Find relevance: Find the place where the two
		marked strings are closest to each other.
		Estimate: Estimate the distance between two
		markers.

# TABLE 2. (Continued.) Task categories in previous research.

Etemadpour et al. [26]	2015	Estimate: Estimate the number of observed cluster/subcluster/outliners. Identify: Identify the closest cluster to a given cluster/object. Rank: Rank the objects based on the distance to a given cluster/ Rank cluster based on density.
Liu et al. [62]	2013	Browsing: Visit the data with no specific goal in mind. Fact finding: Look for specific facts or pieces of information. Information gathering: Collect information often from multiple sources. Users do not always need to know when you have completed the task and there is no specific answer. Revisit: Revisit some source that you previously used.

#### TABLE 3. Category of tasks used in framework.

Task	Description and Examples
Precise	Given characteristic, find something specific (e.g., Find the
Select/	largest amount of sales).
Identify	
Fuzzy	Find something not specified but can be determined by
Select/	professional knowledge (e.g., Find anomaly transactions/
Identify	sale data) or other characteristics (e.g., Find managers that
	has similar performance last year).
Compare	Compare two or more data elements/ sets or distinguish a
	data element/ cluster from others (e.g., Find store manager
	that has a better performance in the past season).
Rank/ Sort	Given a set of data cases, rank them according to some
	ordinal metric (e.g., find top 3 best stores based on sales last
	year).
Filter	Query a set of data by given conditions, may combined with
	other tasks (e.g., find the best store in North Dakota).
Cluster/	Cluster data points that have similar characteristics (e.g.,
segmentat-	Find the average sales for all the stores in a specific area;
ion	Find a state have more sales last year).
Correlate	Find the relationship between two or more elements, include
	trending (e.g., which store has the fastest increase in sales).
Estimate	Based on the information visualized, make an estimate of
	something that is not shown precisely, either because hidden
	data (e.g., estimate how many stores available in North
	Dakota) or data not available (e.g., estimate the total sales
	next year across the country).

2% of statistics major, 4% of business analytics, and rest 4% of plant sciences).

### **B. EXPERIMENTAL DESIGN**

This experiment is a within-group design. The independent variable is design complexity, the combination of interaction types and visual element types. The system has three levels of appropriate design complexity, including increasing complexity in color and interaction. Figure 5 shows a chart selected for one of the designs. Design one has the lowest complexity without toolbars. There is no interaction for this design. It contains minimal visual elements with white/black color and stripe patterns (see Figure 5 (a)). In this design, only two charts have a legend with the toolbars. Design two is of medium complexity. We applied blue colors with different brightness to the chart elements (see Figure 5 (b)).



FIGURE 5. Color selection for different design complexity levels.

Most charts have a legend and simple interactions, such as horizontal movement and one-dimensional zooming, element selection, and reset. Design three has the highest design complexity. We utilized colors with different hue values (see Figure 5 (c)) to classify data and provide a call-out box to display additional information. Similar to design two, the legend is shown in the charts. But the interaction is more complex in design three. Users can perform move and zoom the charts in two dimensions. The toolbar enables the participants to amplify or select the chart elements and toggle call-out boxes. These actions tie into our dependent variables, which involve measuring decision task-based user experience. The dependent variables include decision task-based user experience (mental demand, temporal demand, efforts, performance and frustration level) and the overall user experience (usability, aesthetics, trust, loyalty, emotion involvement, decisionmaking experience and interactivity). These dimensions were measured by SUPR-Q, EIS, DQQ and IPS mentioned above.

# C. MATERIALS AND APPARATUS

We implemented a web-based business sale information visualization system written in HTML and JavaScript for this user study. The sample data are from a US shopping mall dataset, including product sales, refund rate, department turnover, product price and discount change, and the consumption of large-volume buyers, etc. The system contains different visualization types, including treemap, bar chart, bubble chart, and scatter plot, etc. Figure 7 illustrates the design of the user interface. The experiment was conducted on a Windows 10 desktop computer in the human-computer interaction lab. The program was written in HTML and JavaScript and was displayed on a 24-inch LCD monitor with a resolution of 1920  $\times$  1080 pixels.

# D. PROCEDURE

During the experiment, the participants sat straight at the front of the desk. Their eyes were about 1.5 feet away from the screen. In addition, the participants were asked to hold the mouse and operate the program with their dominant hands. Figure 8 shows the process of the user study. The participants were invited into the human-computer interaction lab. After reading and signing the consent form, they were asked to fill in a pre-study questionnaire, followed by a short training familiarizing themselves with the approach and tasks. Then, they completed three tasks on each of three designs. In these

#### TABLE 4. Results of effort degree.

Е	Design 1	Design 2	Design 3	F	р	
Task 1	$2.82 \pm 0.250$	2.16±0.213	1.57±0.151	8.977	0.000	
Task 2	3.25±0.247	2.84±0.219	2.32±0.185	4.558	0.012	
Task 3	3.30±0.282	2.48±0.214	$1.48 \pm 0.119$	17.753	0.000	

tasks, participants typed the answers to the question displayed next to the chart according to their observation (see Figure 6). After each task, the participants were asked to complete an in-study questionnaire to evaluate their experience of that task. Following are the examples of questions for Task1, Task2 and Task3. Task1: In the second half of 2014, which region had the highest sales in September? Task2: In 2014, in which region(s) did component sales surpass bike sales? Task3: In which month of 2015, did the sales of clothing have the highest positive growth rate? After they finished all three tasks for the current design, the participants evaluated their overall impressions of the system with a post-session questionnaire.

# E. RESULT

Among the 47 participants, 3 did not complete all the tasks or answered all the questions. Thus, we collected valid results from 44 participants. All the data collected in our research were analyzed by SPSS 20.0.

#### 1) CORRECTNESS OF TASKS

The scoring metric for tasks are counted as follow: for task 1-3, each correct answer will be recorded as a score 1, incorrect answer will be recorded as score 0. Chi-square analysis was conducted to do cross-tabulation comparisons. The chi-square test indicated there were no significant differences between different designs in task 1 ( $\chi^2 = 0.00$ ,  $\rho = 1.00$ ), task 2 ( $\chi^2 = 3.014$ ,  $\rho = 0.222$ ) and task 3 ( $\chi^2 = 0.266$ ,  $\rho = 0.875$ ).

#### 2) TASK-BASED RATINGS

The results of task rating for three designs on three tasks are analyzed by use of repeated Analysis of Variance (ANOVA). All the detailed statistical analyses for task 1-3 are as follows.

#### a: EFFORT (E) DEGREE

The design was the independent variable, and the rating of effort was the dependent variable. Regard to the effort rating, there were significant differences between the designs among all the tasks (p< 0.05). The result of Least Significant Difference (LSD) Multiple Comparisons showed that: there was significant differences among all designs for task 1 (p < 0.05). there is significant difference between design 1 and 3 for task 2 (p < 0.01), but no significant difference between design 1 and 2 (p > 0.05), design 2 and 3 (p > 0.05). For task 3, there was significant differences among all designs



FIGURE 6. An example of chart for one of three designs.





FIGURE 7. GUI of a single task.

(p < 0.05). Therefore, there was significant differences in general tasks' effort between design 1, 2 and 3.

# b: FRUSTRATION LEVEL (FL) DEGREE

The design was the independent variable, and the frustration level was the dependent variable. Regard to the effort rat-

ing, there were significant differences between the designs among all the tasks (p < 0.001). The result of LSD Multiple Comparisons showed that: For task 1, there was significant differences between design 1 and 2 (p < 0.001), design 1 and 3 (p < 0.001) but no difference between 2 and 3 (p > 0.05).

Next Task

# IEEE Access



FIGURE 8. Procedure of the user study.

For task 2, there was significant differences between design 1 and 3 (p < 0.001), design 2 and 3 (p < 0.01), but no significant difference between design 1 and 2 (p > 0.05). For task 3, there was significant differences among all designs (p < 0.01).

#### TABLE 5. Results of frustration level.

FL	Design 1	Design 2	Design 3	F	р
Task 1	2.89±0.270	1.86±0.191	$1.34 \pm 0.103$	15.417	0.000
Task 2	$2.95 \pm 0.268$	$2.61 \pm 0.240$	$1.66 \pm 0.155$	8.819	0.000
Task 3	3.73±0.249	$3.09 {\pm} 0.237$	$1.82 \pm 0.149$	20.246	0.000

#### c: MENTAL DEMAND (MD) DEGREE

The design was the independent variable, and the mental demand was the dependent variable. Regard to the mental demand, there were significant differences between the designs among all the tasks (p < 0.001). The result of LSD Multiple Comparisons showed that: For task 1, there was significant differences among all designs (p < 0.05). For task 2, there was significant differences between design 1 and 3 (p < 0.001), design 2 and 3 (p < 0.01), but no significant difference between design 1 and 2 (p > 0.05). For task 3, there was significant differences among all designs (p < 0.01).

#### TABLE 6. Results of mental demand degree.

MD	Design 1	Design 2	Design 3	F	р
Task 1	3.23±0.283	$2.48 \pm 0.229$	$1.77 \pm 0.178$	9.674	0.000
Task 2	3.52±0.226	3.32±0.222	$2.39 \pm 0.208$	7.630	0.001
Task 3	4.05±0.262	3.23±0.223	$2.07 \pm 0.161$	20.566	0.000

# d: PERFORMANCE (P) DEGREE

The design was the independent variable, and the performance was the dependent variable. Regard to the effort rating, there were significant differences between the designs among all the tasks (p < 0.001). The result of LSD Multiple Comparisons showed that: For task 1, there was significant differences between design 1 and 2 (p < 0.001), design 1 and 3 (p < 0.001) but no difference between 2 and 3 (p >0.05). For task 2, there was significant differences between design 1 and 3 (p < 0.001), design 2 and 3 (p < 0.01), but no significant difference between design 1 and 2 (p > 0.05). For task 3, there was significant differences among all designs (p < 0.05).

TABLE 7. Results of p	performance degree.
-----------------------	---------------------

Р	Design 1	Design 2	Design 3	F	р
Task 1	6.09±0.172	6.68±0.112	$6.84 \pm 0.086$	9.524	0.000
Task 2	5.66±0.187	5.89±0.173	6.61±0.104	9.886	0.000
Task 3	5.52±0.185	$5.98 \pm 0.164$	6.73±0.081	16.321	0.000

#### e: TEMPORAL DEMAND (TD) DEGREE

The design was the independent variable, and the temporal demand was the dependent variable. Regard to the temporal demand, there were significant differences (p < 0.05) between the designs among all the tasks except task 2 (p > 0.05). The result of LSD Multiple Comparisons showed that: For task 1, there was significant differences between design 1 and 2 (p < 0.05), design 1 and 3 (p < 0.05) but no difference between 2 and 3 (p > 0.05). For task 2, there was significant difference between design 1 and 3 (p < 0.05), but no significant difference between design 1 and 3 (p < 0.05), but no significant difference between design 1 and 2 (p > 0.05), design 2 and 3 (p > 0.05). For task 3, there was significant difference between design 1 and 3 (p < 0.001), but no difference between design 1 and 3 (p < 0.001), but no difference between design 1 and 3 (p < 0.001), but no difference between design 1 and 3 (p < 0.001), but no difference between design 1 and 3 (p < 0.001), but no difference between design 1 and 3 (p < 0.001), but no difference between design 1 and 3 (p < 0.001), but no difference between design 1 and 2 (p > 0.05), design 2 and 3 (p > 0.05).

#### TABLE 8. Results of temporal demand degree.

TD	Design 1	Design 2	Design 3	F	р
Task 1	2.14±0.22	$1.64 \pm 0.149$	$1.52 \pm 0.148$	3.470	0.034
Task 2	$2.30 \pm 0.205$	$2.18 \pm 0.196$	$1.75 \pm 0.157$	2.375	0.097
Task 3	2.55±0.253	$2.07 \pm 0.188$	1.52±0.133	6.736	0.002

#### 3) OVERALL FACTOR ANALYSIS

#### a: ITEM ANALYSIS

The purpose of this analysis was to determine whether the elements were valid and appropriate. First, the distinguishability analysis (T-test) was used to examine whether there was significant difference between high score group (27%) and low score group (73%) on each element. Results showed that all the T-test values were significant (p < 0.001 or p = .001), which meant that a total of 12 elements had a great discrimination and quite reasonable. Second, correlation analysis was applied to calculate the correlation between the total score and each element. Results showed that all the correlations were significant ( $r = 0.410 \sim .924$ , p < 0.01), which indicated all the elements measured the user experience of BI.

# b: EXPLORATORY FACTOR ANALYSIS

The Kaiser-Meyer-Olkin (KMO) analysis and Bartlett's Test of Sphericity showed that the 12 elements were ideal for



FIGURE 9. UX Ratings of three visualization designs.

factor analysis (KMO = 0.942,  $\chi 2 = 1495.301$ , p < 0.001). Table 9 shows the result of exploratory factor analysis.

As illustrated in Table 9, three factors were identified after the exploratory factor analysis. Factor 1 included usability, trust, appearance, loyalty, information quality, and emotional involvement, so Factor 1 was named as Attractiveness. Factor 2 involved decision complexity, auxiliary importance and decision quality, so this factor was called Decision Experience. Factor 3 was similar to the original Interactivity elements, consisting of perceived control, perceived responsiveness, perceived communication, so this factor can be still named as Interactivity. Therefore, the resulting 12 elements constituted three factors: Attractiveness, Decision Experience and Interactivity. Three factors can account for 79.758% total variance.

#### c: RELIABILITY

We also evaluated the internal consistency using Cronbach's  $\alpha$ . The Cronbach's  $\alpha$  we calculated from the samples revealed that three factors had high inner reliability (Cronbach's  $\alpha = 0.712, 0.849, 0.940$ ), and the Cronbach's  $\alpha$  of the total scale of BIV User Experience was 0.721. Since all of them are higher than 0.70, it is showing good internal consistency for the whole questionnaire as well as its three factors.

#### 4) ELEMENT-BASED UX DIFFERENCE ANALYSIS

The results of evaluation rating for different designs on various elements are illustrated in Figure 9. The detailed statistical analyses are as follows.

# a: THE OVERALL EVALUATION DIFFERENCES

Design was the independent variable, and the overall evaluation was the dependent variable. The main effect of design was significant, F (2, 80.236) = 40.082, p < 0.001. The result of LSD Multiple Comparisons among different designs showed that the general evaluation for design 3 was significantly higher than design 1 (p < 0.001) and design 2 (p < 0.001), and the overall evaluation of design 2 were significantly higher than design 1, p > 0.05. Therefore, there were significantly differences in overall evaluation among

#### TABLE 9. Result of exploratory factor analysis.

	Factor 1	Factor 2	Factor 3
Usability(A)	0.860		
Trustiness(A)	0.857		
Appearance(A)	0.825		
Loyalty(A)	0.841		
Information Quality(A)	0.799		
Decision Complexity(B)			0.911
Auxiliary Importance(B)			0.555
Decision Quality(B)			0.704
Perceived Control(C)		0.751	
Perceived Responsiveness(C)		0.729	
Perceived Communication(C)		0.796	
Emotional Involvement(A)	0.657		

#### TABLE 10. Factor based ratings and total ratings.

	Attractiveness	Decision Experience	Interactivity	Total
	7 Hulden veness	Experience	interactivity	Total
Design 1	87.64±5.756	49.45±2.720	38.2±1.476	175.3
Design 2	$108.52 \pm 4.400$	56.66±1.985	40.95±1.096	206.14
Design 3	144.52±3.069	69.32±1.839	45.8±0.996	259.64

these three designs; design 3 performs significantly better than other 2 designs.

#### **b:** THE ATTRACTIVENESS DIFFERENCES

Design was the independent variable, and the attractiveness was the dependent variable. The main effect of design was significant, F (2, 80.746) = 47.889, p < 0.001. The result of LSD Multiple Comparisons among different designs showed the attractiveness of design 1 was significantly lower than design 2 (p < 0.05) and design 3 (p < 0.001), and the attractiveness of design 3 was significantly higher than design 2. Therefore, there were significant differences in attractiveness appraisals among the three designs; design 3 performs significantly better than other 2 designs.

#### c: THE DECISION EXPERIENCE DIFFERENCES

The design was the independent variable, and the decision experience was the dependent variable. The main design was significant, F (2, 84.136) = 21.432, p < 0.001. The result of LSD Multiple Comparisons among different designs showed the decision experience of design 1 was significantly lower than design 3 (p < 0.001), but not significantly lower than design 2 (p > 0.05). The decision experience of design 3 was significantly higher than design 2 (p < 0.001). Therefore, there were significant differences in decision experience among the three designs; design 3 performs significantly better than other 2 designs.

#### d: THE INTERACTIVITY DIFFERENCES

The design as the independent variable and the interactivity was the dependent variable. The main design was significant, F (2, 84.149) = 10.643, p < 0.001. The result of LSD Multiple Comparisons among different designs showed the interactivity of design1 was significantly lower than design 3 (p < 0.001), but not significantly lower than design 2 (p > 0.05). And the interactivity appraisals of design 3 was significantly higher than design 2 (p < 0.001). Therefore, there were significant differences in interactivity among the three designs; design 3 performs significantly better than other 2 designs.

# **V. DISCUSSION**

We have proposed and proved UXBIV framework, a novel evaluation framework that can be used in Business Intelligence Visualization. The aim of this section is to explain rationality, practical application, and future directions of UXBIV framework.

#### A. RATIONALITY OF UXBIV FRAMEWORK

The UXBIV framework is obtained through systematic literature investigation and analysis. Our research team conducted a survey across past research related to UX and business needs. We have investigated several fields, such as business intelligence, visualization, user experience, evaluation and decision-making. Based on literature analysis and professional understanding of this area, the research team constructed an evaluation framework for UX of BIV. In this framework, the definition of each element and potential relationships between elements are based on selected literature. All of these have provided a solid foundation for the rationality of this framework.

Especially, compared with Bačić and Fadlalla [9], the set of independent variables we summarized has more advantages in UX research. Although according to visual mental abilities, Bačić and Fadlalla [9] proposed BIV elements as independent variables such as perception, cognition, memory etc., these elements are too abstract and indirect for BIV designers. For example, if the perception of BIV affects users' decisionmaking performance, how can we through perception quickly identify the corresponding problems of visual design? It is difficult for designers to manipulate. Therefore, in UXBIV, the independent variables should be related to attributes of the system itself or users, including data characteristics, interaction, design, users' background, etc. These independent variables are more practical and easier to manipulate for designers of BIV. Therefore, both for users and designers, UXBIV is a reasonable framework, because it embodies the human centric design philosophy.

Overall user experience, a crucial sub-framework of UXBIV, has good validity and reliability. Firstly, correlation analysis showed that all 12 sub-elements of overall UX certainly measure the user experience of BIV. Secondly, discrimination analysis proved that the 12 sub-elements assumed

in the overall UX can well distinguish BIV designs with different design quality. Thirdly, factor analysis confirmed three main factors of overall UX within the sub-frame we proposed: attractiveness, decision-making experience and interactivity. The statistical results of Cronbach's  $\alpha$  show that the measurement of the overall UX has high stability and reliability.

Information quality was initially considered as a sub-element of decision-making experience, but exploratory factor analysis found that information quality was classified into attractiveness. According to the literature analysis, information quality refers to "relative efficiency to provide the relevant information for the viewer" [94], focusing on whether information quality has a significant impact on decision quality [97], it does not directly reflect the decision quality, but is closer to visual design of interfaces, so it is more reasonable to belong to the category of attractiveness subframe. Figure 10 shows the structure of overall UX based on exploratory factor analysis.

#### **B. PRACTICAL APPLICATION OF UXBIV FRAMEWORK**

Our research fulfills a pressing need in the field of BIV research. First of all, the current research and application of BIV require evaluating UX, but research in this field is still in its infancy. The development of theories and methods in this field lags behind the overall level of current user experience research. Secondly, BI visualization designers are more focusing on research and advancing new visualization technology. BIV evaluation involves theories and methods in psychology, decision science and other fields, but it lacks a systematic evaluation guideline, which has greatly affected the advancement of user experience research in the field of BIV. Thirdly, "user-centered" is an important principle emerged in interaction design in recent years. In order to promote users' perception, analysis and application of business information, BI visualization should also emphasize the "user-centered" principle. Therefore, it is very important and urgent to carry out a series of research on user experience in the field of BIV. The ultimate goal of UXBIV constructed in this research is to gain insight into the user experience of BIV systems, optimize the design of BIV, and promote good understanding, communication and interaction among BIV designers, BIV systems, and BIV users.

As shown in Figure 11, a typical lifecycle of a BIV system includes three elements: BIV users, BIV designers, and the BIV system itself. Users make demands to the designers in order to resolve real world issues and obtain BIV systems. The designers (and developers) work on to provide technical solutions and deliver a BIV system based on user's demand. The built BIV system serves the users and resolves the issues they were facing. By introducing the UXBIV framework, we can benefit all the elements in this lifecycle: This framework is able to evaluate the existing BIV system, collect and evaluate users' feedback, and provide a series of design principles based on UX study in order to guide the designers to improve the BIV system. Obviously, this framework will improve the flow of the lifecycle of BIV system lifecycle.

Although BIV is regarded as the core component of business intelligence [72], only a few papers on BIV user experience have been published. At present, there is no UX framework specially designed for BIV. It is the first time we have put forward a UXBIV framework, which provides valuable insights for the visualization in business fields. The framework provides a powerful tool for systematically collecting various user experiences in detail and refining BIV design principles. The analysis results of this framework can significantly facilitate highlighting the advantages of BIV and improve the competitiveness of BIV development enterprises.

Our framework starts from four main factors: user's backgrounds, data characteristics, interactions and designs. Plus, we also take into consideration characteristic factors and environmental factors. These factors helped us to cover as many independent variables as possible. As a dependent variable, UX analysis includes task-based UX analysis and overall impression UX analysis. With regard to the task-based analysis, we will utilize objective and subjective indicators to evaluate BIV in decision-making tasks. With respect to the overall impression UX analysis, we will assess three main aspects: BIV Attractiveness, Decision-Making Experience and Interactivity. User study design serves as a bridge connecting independent variables and dependent variables. User study design provides task design, method design and research paradigms. Differ from existing research on UX of BIV, this framework evaluates user's experience far more than usability. In UXBIV framework, the most outstanding contribution lies in the category of independent variables, task design, combination of methods and the paradigm, and new elements of dependent variables such as decision experience and interactivity. If you intend to evaluate the UX quality of a BIV design, you should use UXBIV. If you try to compare the UX quality of a set of BIV, you should use UXBIV. If you plan to evaluate the same the BIV design many times, for example, to find out whether the continuous optimization of BIV design has promoted a better user experience, then you should use UXBIV.

The case study was an empirical study that combines behavioral experiment with questionnaire survey. The purpose of this study is to demonstrate how to apply our framework to the practical evaluation of a set of BIV. According to the UXBIV framework, we compare the user experience of three BIV designs with different complexity, following the paradigm in UXBIV. The whole study contained pre-task, in-task, post-task stages. Result analysis consists of three aspects: first, we analyzed the correctness of tasks, which ensured that all the designs in the tests are functional and can finish the objective given. Then we did a task-based analysis with five subjective elements. Subsequently, we analyzed the overall UX from three elements: BIV attractiveness, decision-making experience, and interactivity. According to the results, design 3 with the highest design complexity is significantly rated better than the other two designs in almost



FIGURE 10. The structure of overall UX based on EFA.

all cases on all task-based ratings (mental demand, temporal demand, performance, effort, and frustration) and overall UX (BIV attractiveness, decision-making experience, interactivity). Therefore, appropriately increasing the complexity of visual design, such as color and interaction, can reduce users' workload and pressure, and provide more positive user experience. It shows the sensitivity of our proposed framework to detecting the quality of different designs, and proves that this framework can be used to guide BIV design in the future.

# C. FUTURE DIRECTIONS

With the development of BIV design, the UXBIV framework should be continuously developed and gradually updated. This section points out the future research directions in this field.

First of all, we only provided research on traditional platforms, such as personal computer with mouse, keyboard and monitor. Nowadays there are more and more new technologies that applies to visualizations, such as virtual realities (VR), augmented realities (AR), the metaverse, artificial intelligence (AI), holographs, multi-device visualizations, etc. Our framework may also be adjusted to better detect user experiences of systems with these new technologies.

Secondly, the evaluation of BIV should adopt a combination of various methods. Due to the limitation of our laboratory situation, collection methods in the case study are mostly subjective ratings in questionnaire. Electrophysiology or brain-computer interface (BCI), interview and think-aloud can play an important auxiliary role in discovering various details in system design and understanding, as well as analyzing user experience of BIV.

Thirdly, in our study, most of the participants were students, although many of them were studying in the business field. Considering users' backgrounds, we need to expand the test group to include business professionals who are more familiar and experienced with BI systems. The feedbacks they provide as users can help us further optimize the UXBIV framework.

Finally, in view of environmental factors, such as social backgrounds (culture, economy, etc.) and experimental contexts (online, office, etc.), in order to promote its applicability in diverse environments, UXBIV may also be modified in



FIGURE 11. A typical lifecycle of BIV system.

accordance with changes in environmental factors. This topic is also a meaningful direction in the future.

# **VI. CONCLUSION**

This research is the first to propose a UX framework to investigate the user experience of BIV (UXBIV). This framework is based on rigorous literature survey and analysis. It includes independent variables, dependent variables, research design in BIV evaluation. In this framework, the main contributions are the classification of independent variables, task designs, the combination of methods and the paradigm, and new elements of dependent variables. Moreover, we also undertook a case study to validate this framework and evaluated three BIV designs with different complexities. On the basis of analysis of case study combined with literature survey, we believe that UXBIV is quite a reasonable framework for assessing user experience of BIV. In addition, user experience of three BIV designs is significantly different. Design 3 with the highest design complexity is significantly superior to the other two designs on all task-based rating and overall UX. Therefore, this framework provides a powerful tool for designers to evaluate user experiences of BIV designs. It is expected that UXBIV can promote decision-making performance and customer satisfaction, and enhance the competitiveness and influence of BI development companies.

#### **VII. APPENDIX**

#### A. PRE-STUDY QUESTIONNAIRE EXAMPLE

The following statements/questions ask about participants' information related to our research. Your answers are confidential and are for research purposes only.

- 1. Please tell us your age \_\_\_\_\_
- 2. Gender: (Circle one)
- Male Female Other No Response
- 3. What is your profession?
- Employed with \_\_\_\_\_ years of experience
- □ Student
- $\Box$  Other, please indicate \_\_\_\_

4. If you are a student, please answer the following questions:

Please select: 
Freshman 
Sophomore 
Junior 
Senior
Master student 
Ph.D. student

5. What is your major of study? \_\_\_\_\_

6. How	would y	ou rate	your I	English I	language	skills?
1	2	3	4	5	6	7
Completely Not Proficient	Moderately Not Proficient	Slightly Not Proficient	Neutral	Slightly Proficient	Moderately Proficient	Completely Proficient

Scale for questions 7-9:

1	2	3	4	5	6	7
Extremely	Moderately	Somewhat	Neutral	Somewhat	Moderately	Extremely
Unfamiliar	Unfamiliar	Unfamiliar		Familiar	Familiar	Familiar

7. Please indicate how well you know about business

8. Please indicate how well you know about visualization

9. Please indicate how well you know about business intelligence (BI)

10. Which of the following best describes your frequency of using business intelligence (BI)?

1	2	3	4	5	6	7
Never	Rarely	Occasionally	Moderately	Frequently	Usually	Always

11. Please indicate how well you know about marketing data (sales, inventory, etc.)

1	2	3	4	5	6	7
Extremely	Moderately	Somewhat	Neutral	Somewhat	Moderately	Extremely
Unfamiliar	Unfamiliar	Unfamiliar		Familiar	Familiar	Familiar

# B. IN-STUDY QUESTIONNAIRE EXAMPLE

The following statements/questions ask about your experience while doing task on this BI system. Please respond by checking your choice using the scale ranging from 1 point to 7 point. Your answers are confidential and are for research purposes only.

Scale	for c	uestions	1-4:
-------	-------	----------	------

1	2	3	4	5	6	7
Very Low	Moderately Low	Slightly Low	Neutral	Slightly High	Moderately High	Very High

1. How mentally demanding was required (e.g., thinking, deciding, calculating, remembering, looking, searching, etc.) for the task?

2. How much time pressure did you feel during the task?

3 How hard did you have to work (mentally and physically) to accomplish your level of performance during the task?

4. How discouraged or frustrated did you feel during the task?

5. How successful do you think you were in accomplishing the goals of this task?

1	2	3	4	5	6	7
Very	Moderately	Somewhat	Neutral	Somewhat	Moderately	Very
unsuccessful	unsuccessful	Unsuccessful		Successful	Successful	Successful

# C. POST-STUDY QUESTIONNAIRE EXAMPLE

The following statements/questions ask about your experience while using this BI system. Please respond by checking your choice using the scale ranging from 1 point to 7 point. Your answers are confidential and are for research purposes only.

1) SECTION 1

1	2	3	4	5	6	7
Strongly	Moderately	Somewhat	Neutral	Somewhat	Moderately	Strongly
Disagree	Disagree	Disagree		Agree	Agree	Agree

1. This system is easy to use.

- 2. It is easy to navigate within the system.
- 3. I enjoy using the system.
- 4. I feel comfortable fulfill tasks by using this system.
- 5. I can count on the information I get on this system.
- 6. I found the system to be attractive.
- 7. I feel confident making decisions by using this system.
- 8. The system keeps the promises it makes to me.
- 9. I will likely return to this system in the future.

10. I will recommend this system to peers or colleagues.

# 2) SECTION 2

1	2	3	4	5	6	7
Strongly	Moderately	Somewhat	Neutral	Somewhat	Moderately	Strongly
Disagree	Disagree	Disagree		Agree	Agree	Agree

1. The outcome of the decision depends on the interaction of different factors (variables or elements in the business data such as year, region, sales amount, etc.)

2. The decision involves a large number of factors (variables or elements in the business data such as year, region, sales amount, etc.).

3. I believe I made a good decision.

- 4. The information my BI system provides is:
- 1) Not overwhelming
- 2) Available when I need it
- 3) Easy to extract

5. I relied highly on BI visualization functionality while making the decision.

6. When making the decision I have to consider many different alternatives.

7. How satisfied were you with the decision-making pro-

1	2	3	4	5	6	7
Very	Moderately	Somewhat	Neutral	Somewhat	Moderately	Very
Unsatisfied	Unsatisfied	Unsatisfied		Satisfied	Satisfied	Satisfied

# 3) SECTION 3

1	2	3	4	5	6	7
Strongly	Moderately	Somewhat	Neutral	Somewhat	Moderately	Strongly
Disagree	Disagree	Disagree		Agree	Agree	Agree

1. This system facilitates two-way communication between the users and the system.

(Two-way communication refers to the ability for reciprocal interaction between the system and the user. In such a communication, the system and the user can interact with each other.)

2. The system gives users the opportunity to talk back. (talk back = react, respond)

3. I felt that I had a lot of control over my interactive experiences on this system.

4. While using the system, I could choose freely what I wanted to see.

5. The system processed my input very quickly.

6. Getting information from the system is very fast.

7. I was able to obtain the information I want without any delay.

4) SECTION 4

1	2	3	4	5	6	7
Extremely	Moderately	Slightly	Neutral	Slightly	Moderately	Extremely

1. Boring/Interesting

- 2. Unexciting/Exciting
- 3. Unappealing/Appealing
- 4. Mundane/ fascinating
- 5. Uninvolving /Involving

# ACKNOWLEDGMENT

(Shuqing Liu and Hankuan Zhang contributed equally to this work.)

# REFERENCES

- S. Adikari, C. McDonald, and J. Campbell, "A design science framework for designing and assessing user experience," in *Proc. Int. Conf. Human-Comput. Interact.*, 2011, pp. 25–34, doi: 10.1007/978-3-642-21602-2\_3.
- [2] W. A. W. Adnan, N. G. N. Daud, and N. L. M. Noor, "Expressive information visualization taxonomy for decision support environment," in *Proc.* 3rd Int. Conf. Converg. Hybrid Inf. Technol., Nov. 2008, pp. 88–93, doi: 10.1109/iccit.2008.330.
- [3] M. Ahn, M. Lee, J. Choi, and S. Jun, "A review of brain-computer interface games and an opinion survey from researchers, developers and users," *Sensors*, vol. 14, no. 8, pp. 14601–14633, Aug. 2014, doi: 10.3390/s140814601.

- [4] L. Alben, "Quality of experience: Defining the criteria for effective interaction design," *Interactions*, vol. 3, no. 3, pp. 11–15, May 1996, doi: 10.1145/235008.235010.
- [5] R. A. Amar and J. T. Stasko, "Knowledge precepts for design and evaluation of information visualizations," *IEEE Trans. Vis. Comput. Graphics*, vol. 11, no. 4, pp. 432–442, Jul. 2005, doi: 10.1109/ TVCG.2005.63.
- [6] R. Amar and J. Stasko, "Best paper: A knowledge task-based framework for design and evaluation of information visualizations," in *Proc. IEEE Symp. Inf. Vis.*, Oct. 2004, pp. 143–150.
- [7] D. Ariely, "Controlling the information flow: Effects on consumers' decision making and preferences," *J. Consum. Res.*, vol. 27, no. 2, pp. 233–248, Sep. 2000, doi: 10.1086/314322.
- [8] S. Attar-Khorasani and R. Chalmeta, "Internet of Things data visualization for business intelligence," *Big Data*, Feb. 2022, doi: 10.1089/big.2021.0200.
- [9] D. Bačić and A. Fadlalla, "Business information visualization intellectual contributions: An integrative framework of visualization capabilities and dimensions of visual intelligence," *Decis. Support Syst.*, vol. 89, pp. 77–86, Sep. 2016, doi: 10.1016/j.dss.2016.06.011.
- [10] S. Barlowe, Y. Liu, J. Yang, D. R. Livesay, D. J. Jacobs, J. Mottonen, and D. Verma, "WaveMap: Interactively discovering features from protein flexibility matrices using wavelet-based visual analytics," *Comput. Graph. Forum*, vol. 30, no. 3, pp. 1001–1010, Jun. 2011, doi: 10.1111/j.1467-8659.2011.01949.x.
- [11] L. Bartram, A. Patra, and M. Stone, "Affective color in visualization," in Proc. CHI Conf. Human Factors Comput. Syst., May 2017, pp. 1364–1374, doi: 10.1145/3025453.3026041.
- [12] R. C. Basole, J. Huhtamäki, K. Still, and M. G. Russell, "Visual decision support for business ecosystem analysis," *Expert Syst. Appl.*, vol. 65, pp. 271–282, Dec. 2016, doi: 10.1016/j.eswa.2016.08.041.
- [13] A. Buja, D. Cook, and D. F. Swayne, "Interactive high-dimensional data visualization," *J. Comput. Graph. Statist.*, vol. 5, no. 1, pp. 78–99, 1996. [Online]. Available: http://www.jstor.org/stable/1390754
- [14] N. Cawthon and A. V. Moere, "A conceptual model for evaluating aesthetic effect within the user experience of information visualization," in *Proc.* 10th Int. Conf. Inf. Vis. (IV), 2006, pp. 374–382.
- [15] Y. C. Cho and E. Sagynov, "Exploring factors that affect usefulness, ease of use, trust, and purchase intention in the online environment," *Int. J. Manage. Inf. Syst.*, vol. 19, pp. 21–36, Jan. 2015, doi: 10.19030/ijmis.v19i1.9086.
- [16] W. Chung and A. Leung, "Supporting web searching of business intelligence with information visualization," in *Proc. IEEE/WIC/ACM Int. Conf. Web Intell. (WI)*, Nov. 2007, pp. 807–811, doi: 10.1109/wi. 2007.4427193.
- [17] T. D. Clark and M. C. Jones, "An experimental analysis of the dynamic structure and behavior of managerial support systems," *Syst. Dyn. Rev.*, vol. 24, no. 2, pp. 215–245, Mar. 2008, doi: 10.1002/ sdr.401.
- [18] F. E. Croxton and H. Stein, "Graphic comparisons by bars, squares, circles, and cubes," *J. Amer. Stat. Assoc.*, vol. 27, no. 177, pp. 54–60, Mar. 1932, doi: 10.1080/01621459.1932.10503227.
- [19] D. Angeli, A. Sutcliffe, and J. Hartmann, "Interaction, usability and aesthetics: What influences users' preferences?" in *Proc. 6th Conf. Designing Interactive Syst.*, 2006, pp. 271–280, doi: 10.1145/1142405. 1142446.
- [20] N. Dedi? and C. Stanier, "Measuring the success of changes to existing business intelligence solutions to improve business intelligence reporting," in *Proc. Int. Conf. Res. Practical Issues Enterprise Inf. Syst.*, 2016, pp. 225–236, doi: 10.1007/978-3-319-49944-4\_17.
- [21] W. Didimo, G. Liotta, F. Montecchiani, and P. Palladino, "An advanced network visualization system for financial crime detection," in *Proc. IEEE Pacific Vis. Symp.*, Mar. 2011, pp. 203–210, doi: 10.1109/pacificvis.2011.5742391.
- [22] H. Dudycz, "Usability of business information semantic network search visualization," in *Proc. Multimedia, Interactive, Design Innnovation*, 2015, pp. 1–9, doi: 10.1145/2814464.2814477.
- [23] F. R. Dwyer, P. H. Schurr, and S. Oh, "Developing buyer-seller relationships," *J. Marketing*, vol. 51, no. 2, pp. 11–27, Apr. 1987, doi: 10.1177/002224298705100202.
- [24] O. Embarak, Data Visualization. Cham, Switzerland: Springer, 2018, pp. 293–342, doi: 10.1007/978-1-4842-4109-7\_7.

- [25] M. Eriksson and B. Ferwerda, "Towards a user experience framework for business intelligence," *J. Comput. Inf. Syst.*, vol. 61, no. 5, pp. 428–437, Sep. 2021.
- [26] R. Etemadpour, R. Motta, J. G. D. S. Paiva, R. Minghim, M. C. F. D. Oliveira, and L. Linsen, "Perception-based evaluation of projection methods for multidimensional data visualization," *IEEE Trans. Vis. Comput. Graph.*, vol. 21, no. 1, pp. 81–94, Jan. 2015, doi: 10.1109/TVCG.2014.2330617.
- [27] C. M. N. Faisal, M. Gonzalez-Rodriguez, D. Fernandez-Lanvin, and J. de Andres-Suarez, "Web design attributes in building user trust, satisfaction, and loyalty for a high uncertainty avoidance culture," *IEEE Trans. Hum.-Mach. Syst.*, vol. 47, no. 6, pp. 847–859, Dec. 2017, doi: 10.1109/THMS.2016.2620901.
- [28] A. Figueiras, "Towards the understanding of interaction in information visualization," in *Proc. 19th Int. Conf. Inf. Visualisation*, Jul. 2015, pp. 140–147, doi: 10.1109/iv.2015.34.
- [29] D. Filonik and D. Baur, "Measuring aesthetics for information visualization," in *Proc. 13th Int. Conf. Inf. Vis.*, Jul. 2009, pp. 579–584, doi: 10.1109/iv.2009.94.
- [30] D. J. Finney, "Dimensions of statistics," J. Roy. Stat. Soc., C Appl. Statist., vol. 26, pp. 285–289, Nov. 1977, doi: 10.2307/2346969.
- [31] C. Forsell and M. Cooper, "An introduction and guide to evaluation of visualization techniques through user studies," in *Handbook of Human Centric Visualization*. New York, NY, USA: Springer, 2014, pp. 285–313.
- [32] I. Foudalis, K. Jain, C. Papadimitriou, and M. Sideri, "Modeling social networks through user background and behavior," in *Proc. Int. Workshop Algorithms Models Web-Graph*, 2011, pp. 85–102, doi: 10.1007/978-3-642-21286-4\_8.
- [33] C. M. D. S. Freitas, P. R. G. Luzzardi, R. A. Cava, M. Winckler, M. S. Pimenta, and L. P. Nedel, "On evaluating information visualization techniques," in *Proc. Work. Conf. Adv. Vis. Interface*, May 2002, pp. 373–374, doi: 10.1145/1556262.1556326.
- [34] B. Fu, N. F. Noy, and M.-A. Storey, "Eye tracking the user experience—An evaluation of ontology visualization techniques," *Semantic Web*, vol. 8, no. 1, pp. 23–41, Nov. 2016.
- [35] E. Y. Gorodov and V. V. Gubarev, "Analytical review of data visualization methods in application to big data," *J. Electr. Comput. Eng.*, vol. 2013, 2013, Art. no. 969458, doi: 10.1155/2013/969458.
- [36] M. S. Gounder, V. V. Iyer, and A. Al Mazyad, "A survey on business intelligence tools for university dashboard development," in *Proc. 3rd MEC Int. Conf. Big Data Smart City (ICBDSC)*, Mar. 2016, pp. 1–7, doi: 10.1109/icbdsc.2016.7460347.
- [37] S. G. Hart, "NASA task load index," NASA Ames Res. Center, Moffett Field, CA, USA, 1986. Accessed: Jun. 21, 2022. [Online]. Available: https://ntrs.nasa.gov/citations/20000021487
- [38] M. Hassenzahl, S. Diefenbach, and A. Göritz, "Needs, affect, and interactive products—Facets of user experience," *Interacting Comput.*, vol. 22, no. 5, pp. 353–362, Sep. 2010, doi: 10.1016/j.intcom.2010.04.002.
- [39] J. Heer, N. Kong, and M. Agrawala, "Sizing the horizon: The effects of chart size and layering on the graphical perception of time series visualizations," in *Proc. SIGCHI Conf. Human Factors Comput. Syst.*, Apr. 2009, pp. 1303–1312, doi: 10.1145/1518701.1518897.
- [40] J. G. Hollands and I. Spence, "Judgments of change and proportion in graphical perception," *Hum. Factors, J. Hum. Factors Ergonom. Soc.*, vol. 34, no. 3, pp. 313–334, Jun. 1992, doi: 10.1177/001872089203400306.
- [41] H. Hrimech, L. Alem, and F. Merienne, "How 3D interaction metaphors affect user experience in collaborative virtual environment," *Adv. Hum.-Comput. Interact.*, vol. 2011, Sep. 2011, Art. no. 172318, doi: 10.1155/2011/172318.
- [42] M. I. Hwang, "Decision making under time pressure: A model for information systems research," *Inf. Manage.*, vol. 27, no. 4, pp. 197–203, Oct. 1994, doi: 10.1016/0378-7206(94)90048-5.
- [43] V. Imamoglu, "Social background and user response," *Batiment Int.*, *Building Res. Pract.*, vol. 13, no. 4, pp. 243–247, Jul. 1985, doi: 10.1080/09613218508551218.
- [44] Ergonomics of Human-System Interaction, ISO Standard 9241-210, 2010.
- [45] H. Jang and S. H. Han, "User experience framework for understanding user experience in blockchain services," *Int. J. Hum.-Comput. Stud.*, vol. 158, Feb. 2022, Art. no. 102733, doi: 10.1016/j.ijhcs.2021.102733.
- [46] Z. Jiang, J. Chan, B. Tan, and W. Chua, "Effects of interactivity on website involvement and purchase intention," *J. Assoc. Inf. Syst.*, vol. 11, no. 1, pp. 34–59, Jan. 2010, doi: 10.17705/1jais.00218.

- [47] J.-Y. Jian, A. M. Bisantz, and C. G. Drury, "Foundations for an empirically determined scale of trust in automated systems," *Int. J. Cognit. Ergonom.*, vol. 4, no. 1, pp. 53–71, Mar. 2000, doi: 10.1207/ s15327566ijce0401\_04.
- [48] G. J. Johnson, G. C. Bruner, and A. Kumar, "Interactivity and its facets revisited: Theory and empirical test," *J. Advertising*, vol. 35, no. 4, pp. 35–52, Dec. 2006, doi: 10.2753/joa0091-3367350403.
- [49] M. Khan and S. S. Khan, "Data and information visualization methods, and interactive mechanisms: A survey," *Int. J. Comput. Appl.*, vol. 34, pp. 1–14, Dec. 2011.
- [50] S. M. Khan, M. Rahman, A. Apon, and M. Chowdhury, "Characteristics of intelligent transportation systems and its relationship with data analytics," in *Data Analytics for Intelligent Transportation Systems*. Amsterdam, The Netherlands: Elsevier, 2017, pp. 1–29, doi: 10.1016/b978-0-12-809715-1.00001-8.
- [51] M. A. Khawaja, F. Chen, and N. Marcus, "Measuring cognitive load using linguistic features: Implications for usability evaluation and adaptive interaction design," *Int. J. Hum.-Comput. Interact.*, vol. 30, no. 5, pp. 343–368, May 2014, doi: 10.1080/10447318. 2013.860579.
- [52] M. C. Kim, Y. Zhu, and C. Chen, "How are they different? A quantitative domain comparison of information visualization and data visualization (2000–2014)," *Scientometrics*, vol. 107, no. 1, pp. 123–165, Apr. 2016, doi: 10.1007/s11192-015-1830-0.
- [53] D. Zakay, "Decision making in action: Models and methods," Acta Psychologica, vol. 85, no. 3, pp. 263–264, May 1994.
- [54] S. Koshman, "Testing user interaction with a prototype visualization-based information retrieval system," *J. Amer. Soc. Inf. Sci. Technol.*, vol. 56, no. 8, pp. 824–833, 2005, doi: 10.1002/asi.20175.
- [55] S. Kremer and U. Lindemann, "A framework for understanding, communicating and evaluating user experience potentials," in *Proc. 20th Int. Conf. Eng. Design (ICED)*, 2015, p. S-517.
- [56] R. van Lammeren, J. Houtkamp, S. Colijn, M. Hilferink, and A. Bouwman, "Affective appraisal of 3D land use visualization," *Comput., Environ. Urban Syst.*, vol. 34, no. 6, pp. 465–475, Nov. 2010, doi: 10.1016/j.compenvurbsys.2010.07.001.
- [57] E. L.-C. Law, V. Roto, M. Hassenzahl, Arnold, and J. Kort, "Understanding, scoping and defining user experience: A survey approach," in *Proc. SIGCHI Conf. Human Factors Comput. Syst.*, 2009, pp. 719–728, doi: 10.1145/1518701.1518813.
- [58] B. Lee, C. Plaisant, C. S. Parr, J.-D. Fekete, and N. Henry, "Task taxonomy for graph visualization," in *Proc. AVI Workshop BEyond Time Errors, Novel Eval. Methods Inf. Vis.*, 2006, pp. 1–5, doi: 10.1145/1168149.1168168.
- [59] J. D. Lee and K. A. See, "Trust in automation: Designing for appropriate reliance," *Hum. Factors, J. Hum. Factors Ergonom. Soc.*, vol. 46, no. 1, pp. 50–80, 2004, doi: 10.1518/hfes.46.1.50.30392.
- [60] J. R. Lewis, "Psychometric evaluation of an after-scenario questionnaire for computer usability studies: The ASQ," ACM SIGCHI Bull., vol. 23, no. 1, pp. 78–81, Jan. 1991, doi: 10.1145/122672.122692.
- [61] Y. Liu, A.-L. Osvalder, and S. Dahlman, "Exploring user background settings in cognitive walkthrough evaluation of medical prototype interfaces: A case study," *Int. J. Ind. Ergonom.*, vol. 35, no. 4, pp. 379–390, Apr. 2005, doi: 10.1016/j.ergon.2004.10.004.
- [62] Y. Liu, S. Barlowe, Y. Feng, J. Yang, and M. Jiang, "Evaluating exploratory visualization systems: A user study on how clustering-based visualization systems support information seeking from large document collections," *Inf. Vis.*, vol. 12, no. 1, pp. 25–43, Jan. 2013.
- [63] Y. Liu, "Developing a scale to measure the interactivity of websites," J. Advertising Res., vol. 43, no. 2, pp. 207–216, Jun. 2003, doi: 10.2501/jar-43-2-207-216.
- [64] Y. Liu and L. J. Shrum, "What is interactivity and is it always such a good thing? Implications of definition, person, and situation for the influence of interactivity on advertising effectiveness," J. Advertising, vol. 31, no. 4, pp. 53–64, Dec. 2002, doi: 10.1080/00913367. 2002.10673685.
- [65] M. A. Livingston and J. W. Decker, "Evaluation of multivariate visualizations: A case study of refinements and user experience," *Proc. SPIE*, vol. 8294, Jan. 2012, Art. no. 82940G, doi: 10.1117/12.912192.
- [66] H. Ltifi, E. Benmohamed, C. Kolski, and M. Ben Ayed, "Adapted visual analytics process for intelligent decision-making: Application in a medical context," *Int. J. Inf. Technol. Decis. Making*, vol. 19, no. 01, pp. 241–282, Jan. 2020, doi: 10.1142/s0219622019500470.

- [67] M. Maguire, "Methods to support human-centred design," Int. J. Hum.-Comput. Stud., vol. 55, no. 4, pp. 587–634, Oct. 2001, doi: 10.1006/ijhc.2001.0503.
- [68] C. R. McCarthy, "Historical background of clinical trials involving women and minorities," *Academic Med.*, vol. 69, pp. 695–698, Sep. 1994, doi: 10.1097/00001888-199409000-00002.
- [69] S. J. McMillan and J.-S. Hwang, "Measures of perceived interactivity: An exploration of the role of direction of communication, user control, and time in shaping perceptions of interactivity," *J. Advertising*, vol. 31, no. 3, pp. 29–42, Oct. 2002, doi: 10.1080/00913367. 2002.10673674.
- [70] T. Merčun, "Evaluation of information visualization techniques: Analysing user experience with reaction cards," in *Proc. 5th Workshop Beyond Time Errors, Novel Eval. Methods Vis.*, Nov. 2014, pp. 103–109, doi: 10.1145/2669557.2669565.
- [71] A. V. Moere, M. Tomitsch, C. Wimmer, B. Christoph, and T. Grechenig, "Evaluating the effect of style in information visualization," *IEEE Trans. Vis. Comput. Graph.*, vol. 18, no. 12, pp. 2739–2748, Dec. 2012, doi: 10.1109/TVCG.2012.221.
- [72] D. A. Mohan, "Big data analytics: Recent achievements and new challenges," *Int. J. Comput. Appl. Technol. Res.*, vol. 5, no. 7, pp. 460–464, Jul. 2016, doi: 10.7753/ijcatr0507.1008.
- [73] W. Müller and H. Schumann, "Visualization for modeling and simulation: Visualization methods for time-dependent data-an overview," in *Proc. 35th Conf. Winter Simul., Driving Innov.*, 2003, pp. 737–745.
- [74] N. Gershon and S. G. Eick, "Information visualization applications in the real world," *IEEE Comput. Graph. Appl.*, vol. 17, no. 4, p. 66a, Jul. 1997.
- [75] D. Norman, J. Miller, and A. Henderson, "What you see, some of what's in the future, and how we go about doing it: HI at Apple Computer," in *Proc. Conf. Companion Human Factors Comput. Syst.*, 1995, p. 155, doi: 10.1145/223355.223477.
- [76] A. Nowak, D. Karastoyanova, F. Leymann, A. Rapoport, and D. Schumm, "Flexible information design for business process visualizations," in *Proc.* 5th IEEE Int. Conf. Service-Oriented Comput. Appl. (SOCA), Dec. 2012, pp. 1–8, doi: 10.1109/SOCA.2012.6449436.
- [77] S. Ntoa, G. Margetis, M. Antona, and C. Stephanidis, "User experience evaluation in intelligent environments: A comprehensive framework," *Technologies*, vol. 9, no. 2, p. 41, May 2021, doi: 10.3390/technologies9020041.
- [78] T. Olsson, "Concepts and subjective measures for evaluating user experience of mobile augmented reality services," in *Human Factors in Augmented Reality Environments*. Cham, Switzerland: Springer, 2013, pp. 203–232.
- [79] H. Park, M. A. Bellamy, and R. C. Basole, "Visual analytics for supply network management: System design and evaluation," *Decis. Support Syst.*, vol. 91, pp. 89–102, Nov. 2016, doi: 10.1016/j.dss. 2016.08.003.
- [80] R. M. Pillat, E. R. A. Valiati, and C. M. D. S. Freitas, "Experimental study on evaluation of multidimensional information visualization techniques," in *Proc. Latin Amer. Conf. Human-Comput. Interact. (CLIHC)*, 2005, pp. 20–30, doi: 10.1145/1111360.1111363.
- [81] R. S. Raghav, S. Pothula, T. Vengattaraman, and D. Ponnurangam, "A survey of data visualization tools for analyzing large volume of data in big data platform," in *Proc. Int. Conf. Commun. Electron. Syst. (ICCES)*, Oct. 2016, pp. 1–6, doi: 10.1109/cesys.2016.7889976.
- [82] A. Rind, W. Aigner, M. Wagner, S. Miksch, and T. Lammarsch, "Task cube: A three-dimensional conceptual space of user tasks in visualization design and evaluation," *Inf. Vis.*, vol. 15, no. 4, pp. 288–300, Oct. 2016, doi: 10.1177/1473871615621602.
- [83] B. Saket, A. Endert, and J. Stasko, "Beyond usability and performance: A review of user experience-focused evaluations in visualization," in *Proc. 6th Workshop Beyond Time Errors Novel Eval. Methods Vis.*, vol. 2016, pp. 133–142, doi: 10.1145/2993901.2993903.
- [84] A. Satyanarayan, K. Wongsuphasawat, and J. Heer, "Declarative interaction design for data visualization," in *Proc. 27th Annu. ACM Symp. User Interface Softw. Technol.*, 2014, pp. 669–678, doi: 10.1145/2642918.2647360.
- [85] J. Sauro, "SUPR-Q: A comprehensive measure of the quality of the website user experience," J. Usability Stud., vol. 10, no. 2, pp. 68–86, 2015.
- [86] J. H. Song and G. M. Zinkhan, "Determinants of perceived web site interactivity," *J. Marketing*, vol. 72, no. 2, pp. 99–113, Mar. 2008, doi: 10.1509/jmkg.72.2.99.

- [87] I. Spence, "The apparent and effective dimensionality of representations of objects," *Hum. Factors, J. Hum. Factors Ergonom. Soc.*, vol. 46, no. 4, pp. 738–747, Dec. 2004, doi: 10.1518/hfes.46. 4.738.56809.
- [88] J. Stasko, R. Catrambone, M. Guzdial, and K. Mcdonald, "An evaluation of space-filling information visualizations for depicting hierarchical structures," *Int. J. Hum.-Comput. Stud.*, vol. 53, no. 5, pp. 663–694, Nov. 2000, doi: 10.1006/ijhc.2000.0420.
- [89] D. P. Tegarden, "Business information visualization," Commun. Assoc. Inf. Syst., vol. 1, no. 1, p. 4, 1999, doi: 10.17705/1cais.00104.
- [90] A. Thayer and T. E. Dugan, "Achieving design enlightenment: Defining a new user experience measurement framework," in *Proc. IEEE Int. Prof. Commun. Conf.*, Jul. 2009, pp. 1–10, doi: 10.1109/IPCC. 2009.5208681.
- [91] D. Timmermans and C. Vlek, "Multi-attribute decision support and complexity: An evaluation and process analysis of aided versus unaided decision making," *Acta Psychologica*, vol. 80, pp. 49–65, Aug. 1992, doi: https://doi.org/10.1016/0001-6918(92)90040-k.
- [92] E. Torres-Moraga, A. Z. Vásquez-Parraga, and J. Zamora-González, "Customer satisfaction and loyalty: Start with the product, culminate with the brand," *J. Consum. Marketing*, vol. 25, no. 5, pp. 302–313, Aug. 2008, doi: 10.1108/07363760810890534.
- [93] N. Tractinsky, "Toward the study of aesthetics in information technology," in Proc. Int. Conf. Inf. Syst., 2004, pp. 1–10.
- [94] N. Tractinsky and J. Meyer, "Chartjunk or goldgraph? Effects of presentation objectives and content desirability on information presentation," *MIS Quart.*, vol. 23, pp. 397–420, Sep. 1999, doi: 10.2307/ 249469.
- [95] G. H. van Bruggen, A. Smidts, and B. Wierenga, "Improving decision making by means of a marketing decision support system," *Manage. Sci.*, vol. 44, no. 5, pp. 645–658, May 1998, doi: 10.1287/ mnsc.44.5.645.
- [96] A. P. O. S. Vermeeren, L.-C. E. Law, V. Roto, M. Obrist, J. Hoonhout, and K. Väänänen, "User experience evaluation methods: Current state and development needs," in *Proc. 6th Nordic Conf. Human-Comput. Interact., Extending Boundaries*, 2010, pp. 521–530, doi: 10.1145/1868914.1868973.
- [97] L. L. Visinescu, M. C. Jones, and A. Sidorova, "Improving decision quality: The role of business intelligence," *J. Comput. Inf. Syst.*, vol. 57, no. 1, pp. 58–66, Jan. 2017.
- [98] C. Wang, J. Gao, L. Li, and H.-W. Shen, "A multiresolution, volume rendering framework for large-scale time-varying data visualization," in *Proc. 4th Int. Workshop Volume Graph.*, 2005, pp. 11–223, doi: 10.1109/vg.2005.194092.
- [99] W. Wirth, M. Hofer, and H. Schramm, "The role of emotional involvement and trait absorption in the formation of spatial presence," *Media Psychol.*, vol. 15, no. 1, pp. 19–43, Mar. 2012, doi: 10.1080/15213269. 2011.648536.
- [100] P. Wright, J. Wallace, and J. McCarthy, "Aesthetics and experiencecentered design," ACM Trans. Comput.-Hum. Interact., vol. 15, no. 4, pp. 1–21, Nov. 2008, doi: 10.1145/1460355.1460360.
- [101] J. S. Yi, Y. A. Kang, J. Stasko, and J. A. Jacko, "Toward a deeper understanding of the role of interaction in information visualization," *IEEE Trans. Vis. Comput. Graph.*, vol. 13, no. 6, pp. 1224–1231, Nov. 2007, doi: 10.1109/TVCG.2007.70515.
- [102] Y. Yun, D. Ma, and M. Yang, "Human-computer interactionbased decision support system with applications in data mining," *Future Gener. Comput. Syst.*, vol. 114, pp. 285–289, Jan. 2021, doi: 10.1016/j.future.2020.07.048.
- [103] Y. Yu and X. Wu, "Research on user experience design of B2C E-commerce based on personalized needs," in *Proc. 2nd Int. Conf. Intell. Human-Mach. Syst. Cybern.*, Aug. 2010, pp. 100–103, doi: 10.1109/ihmsc.2010.124.
- [104] J. L. Zaichkowsky, "The personal involvement inventory: Reduction, revision, and application to advertising," *J. Advertising*, vol. 23, no. 4, pp. 59–70, Dec. 1994, doi: 10.1080/00913367.1943.10673459.
- [105] M. Zarour and M. Alharbi, "User experience framework that combines aspects, dimensions, and measurement methods," *Cogent Eng.*, vol. 4, no. 1, Jan. 2017, Art. no. 1421006, doi: 10.1080/23311916. 2017.1421006.
- [106] H. Zhang, "An evaluation framework for business intelligence visualization," M.S. thesis, Dept. Comput. Sci., North Dakota State Univ., Fargo, ND, USA, 2022.

- [107] R. Magdalena, Y. Ruldeviyani, D. I. Sensuse, and C. Bernando, "Methods to enhance the utilization of business intelligence dashboard by integration of evaluation and user testing," in *Proc. 3rd Int. Conf. Informat. Comput. Sci. (ICICoS)*, Oct. 2019, pp. 1–6, doi: 10.1109/icicos48119.2019.8982481.
- [108] A. Muppidi, A. S. B. Hashim, and M. H. B. Hasan, "Proposed userexperience model for the design and development of BI dashboards," in *Proc. 2nd Int. Conf. Intell. Cybern. Technol. Appl. (ICICyTA)*, Dec. 2022, pp. 23–28, doi: 10.1109/icicyta57421.2022.10037904.
- [109] S. Almasi, K. Bahaadinbeigy, H. Ahmadi, S. Sohrabei, and R. Rabiei, "Usability evaluation of dashboards: A systematic literature review of tools," *BioMed Res. Int.*, vol. 2023, Feb. 2023, Art. no. 9990933, doi: 10.1155/2023/9990933.
- [110] T. M. Orlando and W. D. Sunindyo, "Designing dashboard visualization for heterogeneous stakeholders (case study: ITB central library)," in *Proc. Int. Conf. Data Softw. Eng. (ICoDSE)*, Nov. 2017, pp. 1–6, doi: 10.1109/ICODSE.2017.8285872.
- [111] C. Amyrotos, "Adaptive visualizations for enhanced data understanding and interpretation," in *Proc. 29th ACM Conf. User Model., Adaptation Personalization*, Jun. 2021, pp. 291–297, doi: 10.1145/3450613.3459657.
- [112] C. Burnay, S. Bouraga, S. Faulkner, and I. Jureta, "User-experience in business intelligence—A quality construct and model to design supportive BI dashboards," in *Research Challenges in Information Science* (Lecture Notes in Business Information Processing), vol. 385. Cham, Switzerland: Springer, 2020, pp. 174–190, doi: 10.1007/978-3-030-50316-1\_11.



**SHUQING LIU** (Member, IEEE) received the bachelor's degree in psychology from Central China Normal University, the M.Sc. degree in cognitive psychology from the Chinese Academy of Sciences, and the Ph.D. degree in sociology from Tsinghua University, China. She is currently a Faculty Member of Tianjin University. Her research interests include user experience and design psychology, consumer behavior, human behaviors in virtual reality, business intelligence visualization.

media psychology, and cultural psychology.



HANKUAN ZHANG received the B.S. degree in electrical engineering and the M.S. degree in software engineering from North Dakota State University, Fargo, ND, USA, in 2015 and 2022, respectively. From 2016 to 2019, he was a Teaching Assistant with North Dakota State University. Since 2020, he has been a Software Engineer with JLL Technology Company Ltd., San Francisco, CA, USA, where he is involved in human-computer interaction and user's experience.



**ZHENHUA YANG** received the M.Des. degree in user-centered design from the University of Cincinnati, the M.F.A. degree in packaging design from the Hunan University of Technology, and the Ph.D. degree in software and security engineering from North Dakota State University. He is currently an Assistant Professor with the Department of Graphic Design, Iowa State University. His research interests include user experience design, human–computer interaction, usable security, and

design for use behavioral change.

# **IEEE**Access



**JUN KONG** received the B.S. degree in computer science from the Huazhong University of Science and Technology, in 1998, the M.S. degree in computer science from Shanghai Jiao Tong University, in 2001, and the Ph.D. degree in computer science from The University of Texas at Dallas, in 2005. He has been a Professor of computer science with North Dakota State University, Fargo, ND, USA. His research and teaching interests include human–computer interaction, software engineer-

ing, and brain-computer interface.



**LIMIN ZHANG** received the Ph.D. degree in management information systems (MIS) from The University of Arizona. She is currently an Associate Professor with the Department of Accounting and Information Systems, College of Business, North Dakota State University. She has taught a variety of MIS courses, including principles of MIS, database management, applied business intelligence, big data analytics, and information management. Her research interests include social

media analytics, financial text mining, business data visualization, and business use of artificial intelligence.



**CHUNHAI GAO** received the bachelor's degree in psychology from Leshan Normal University and the M.Sc. and Ph.D. degrees in development and educational psychology from Southwest University. He is currently an Assistant Professor with the Faculty of Education, Shenzhen University. His research interests include product design and evaluation in education and business, intelligent education and marketing, and child development and education.

. . .